\documentclass[12pt]{article}
\topmargin-.5in
\textheight9in
\oddsidemargin0in
\textwidth6.5in
\parindent= 0.5in
\begin{document}
\centerline{\bf{FERPA WAIVER: WRITTEN HOMEWORK}}
I, the undersigned, relinquish my FERPA rights in the following context: My written
homework for Math 23, Fall 2018, may be returned en masse with others in the class
via the homework boxes in Kemeny Hall. I understand that my homework scores may
be visible to others.
If I choose not to relinquish my FERPA rights, I understand that I will have to present
my student ID at my instructor’s office in order to retrieve my homework sets.

\vspace{.8in}
\noindent
Name (print):$\rule{10cm}{.15mm}$

\vspace{.8in}
\noindent
Name (signature):$\rule{10cm}{.15mm}$
\end{document}

